Author's response to reviews

Title: Screening for known mutations in EIF2B genes in a large panel of patients with premature ovarian failure

Authors:
Anne Fogli (anne.fogli@inserm.u-clermont1.fr)
Fernande Gauthier-Barichard (fernande.gauthier@inserm.u-clermont1.fr)
Raphael Schiffmann (RS4e@nih.gov)
Vien H Vanderhoof (vanderhv@mail.nih.gov)
Vladimir K Bakalov (bakalov@mail.nih.gov)
Lawrence M Nelson (nelsonl@cc1.nichd.nih.gov)
Odile Boespflug-Tanguy (odile.boespflug@inserm.u-clermont1.fr)

Version: 3 Date: 19 October 2004

Author's response to reviews:

Dear Editor,

Thank you for your letter of October 18, 2004.

As suggested, we corrected our manuscript in the following manner:

1. We removed all unnecessary capitalisation from the author list. The term EIF2B (in the title and in the text) must be written with capital letters as it concerns genes.
2. We corrected the English mistakes in the abstract and in the manuscript: "enzymatic restrictions" in "restriction enzyme". We re-wrote the results section in the abstract.
3. We used the template to format the reference list in the journal's style.

We hope that this manuscript is now in a form acceptable for publication in the BMC Women's Health.

Sincerely yours.

Professor Odile BOESPFLUG-TANGUY, Corresponding author
Phone: 33 473 44 86 57 Fax: 33 473 27 74 94
Email: Odile.Boespflug@inserm.u-clermont1.fr